



UNSW
A U S T R A L I A

Rachel Grove
School of Psychiatry
University of New South Wales, NSW Australia 2052
Phone +61 (0)416 242 162
Email r.grove@unsw.edu.au

29th May 2017

Please see attached a revised publication in the *World Journal of Psychiatry* the enclosed manuscript " Transition to School for Children with Autism Spectrum Disorder: A Systematic Review" by Annabel Marsh, Vanessa Spagnol, Rachel Grove and Valsamma Eapen.

The manuscript has not been previously published and has not been submitted elsewhere. All contributors have read and approved the final version of the manuscript.

Sincerely yours,

Valsamma Eapen